Author's response to reviews

Title: Uroflowmetry Nomogram in Iranian Children aged 7 to 14 years

Authors:
Abdol-Mohammad Kajbafzadeh (kajbafzd@sina.tums.ac.ir)
Cyrus Ahmadi Yazdi (cyrus_ahmadi@hotmail.com)
Omid Rouhi (omid_rouhi@yahoo.com)
Parvin Mohseni (parvinmohseni@yahoo.com)

Version: 3 Date: 6 December 2004

Author's response to reviews: see over
Esteemed members of the Editorial Board,

Enclosed please find the manuscript "MS 1099472903373329: Uroflowmetry Nomogram in Iranian Children aged 7 to 14 Years" that we are resubmitting to BMC Urology for consideration.

The esteemed reviewer requires the hypotheses to which the p-values of the results section refer. I should like to attract your kind attention to point that the p-values give the probability that the regression line slope (R: regression coefficient) is zero, which would indicate no correlation between the two variables.

Thank you in advance for your consideration and timely re-review of this manuscript and wishing you a VERY HAPPY NEW YEAR.

Sincerely Yours,

A.M. Kajbafzadeh, MD
Department of Pediatric Urology,
Tehran University of Medical Sciences